Author's response to reviews

Title: The sense of coherence and styles of success in the medical career: a longitudinal study

Authors:

Malgorzata Tartas (mtartas@gumed.edu.pl)
Maciej Walkiewicz (walkiewicz@gumed.edu.pl)
Waldemar Budzinski (wbudzin@gumed.edu.pl)
Mikolaj Majkowicz (mmajk@gumed.edu.pl)
Krzysztof Wojcikiewicz (kwojcikiewicz@wss.gda.pl)

Version: 6 Date: 13 October 2014

Author's response to reviews:

Dear Dr Gabrielle Finn,

Thank you for your kind letter and the offer for our paper to be published on BMC Medical Education.

We are grateful for the Board's decision and highly interested to take up your offer.

We have revised our manuscript taking under the consideration your comments.
1. We have added brief description of the context of the study in the Abstract section.
2. We have added ethics information in the Methods (Participants) section.
3. We have placed information about funding into the Acknowledgements section.
4. We have embed the references into the main text.
5. Finally we have included a figure legend according to the journal guidelines at the end of the document, after the references.

Looking for hearings from you

Yours faithfully

Dr Malgorzata Tartas
Department of Psychology, Medical University of Gdansk, Poland